Author's response to reviews

Title: Narrative Exposure Therapy for immigrant children traumatized by war: study protocol for a randomized controlled trial of effectiveness and mechanisms of change

Authors:
Samuli Kangaslampi (kangaslampi.h.samuli@student.uta.fi)
Ferdinand Garoff (ferdinand.garoff@gmail.com)
Kirsi Peltonen (kirsi.peltonen@uta.fi)

Version: 3 Date: 4 June 2015

Author's response to reviews:

The requested documents on ethical approval and external funding approval have been forwarded to BMCSeriesEditorial@biomedcentral.com.

The following changes and additions have been made to the manuscript:

- As requested, the date of trial registration has been added to the abstract and to the main body of the manuscript under "Ethical issues"

- The section of the manuscript "Ethical issues" has been revised to give more specific details on the ethical approval provided by the Regional Ethics Committee of Tampere University Hospital